## SPECIAL REPORT

## THE ROLE OF BOTULINUM TOXIN IN MULTIMODAL TREATMENT OF SPASTICITY IN AMBULATORY CHILDREN WITH SPASTIC CEREBRAL PALSY: EXTENSIVE EVALUATION OF A COST-EFFECTIVENESS TRIAL\*

Johannes B. J. BUSSMANN, PhD<sup>1</sup>, Robert F. PANGALILA, MD, PhD<sup>2</sup>, Henk J. STAM, MD, PhD, FRCP<sup>1</sup> and Fabienne C. SCHASFOORT, PhD<sup>1,2</sup>

*From the <sup>1</sup>Department of Rehabilitation Medicine, Erasmus MC University Medical Centre and*<sup>2</sup>*Rijndam Rehabilitation, Rotterdam, The Netherlands* 

**Background:** A cost-effectiveness trial (the Space Bop study) on the added value of botulinum toxin injections (BoNT-A) in the leg muscles, as part of a multimodal intervention for ambulatory children with spastic cerebral palsy in the context of a single distinct cycle of care was performed recently by our group. For a broad set of effect outcomes, we found that BoNT-A had no added value if children received comprehensive rehabilitation. However, this counterintuitive finding was met with scepticism.

**Objective:** Since several noteworthy facts and experiences were recorded during the course of the trial and the dissemination phase, the aim of this paper was to describe and discuss some crucial aspects of, and barriers to, the Space Bop study, related to context and perspective, design and results, as well as publication and implementation.

Methods: This paper discusses 5 issues: (i) the design, interpretation and presentation of previous research; (ii) the role of one's own clinical experience and interpretation; (iii) the aims of (BoNT-A) treatment; (iv) conflict of interest, role of industry, and the role of history; (v) optimal treatment modalities and dose-response relationships.

*Conclusion:* Despite the unambiguous findings from the Space Bop study, several factors hindered acceptance of the results. Awareness of these factors is important when performing rehabilitation research and disseminating and implementing research findings.

Key words: cerebral palsy; methodology; evaluation; botulinum toxin; spasticity; physiotherapy, casting, orthosis.

Accepted Apr 3, 2020; Epub ahead of print Apr 16, 2020

J Rehabil Med 2020: 52: jrm00059

Correspondence address: Fabienne Schasfoort, Department of Rehabilitation Medicine, Erasmus MC, University Medical Centre Rotterdam, PO Box 2040, NL-3000 CA Rotterdam, The Netherlands. E-mail: f.schasfoort@erasmusmc.nl

In ambulatory children with spastic cerebral palsy (CP) in the lower limbs, repeated treatment with botulinum toxin injections (BoNT-A) combined with physiotherapy, casting and/or orthosis as adjunctive rehabilitation interventions is a well-established treatment paradigm. There

## LAY ABSTRACT

Recently, we performed a study on the added value of botulinum toxin injections (BoNT-A) for ambulatory children with spastic cerebral palsy (the Space Bop study). The results showed that BoNT-A did not add any value in this context, but this counterintuitive finding was met with scepticism. The aim of this paper is to describe and discuss some crucial aspects of, and barriers to, the Space Bop study. The paper also discusses the design, interpretation and presentation of previous research; the role of clinician's own experience and interpretation; the aims of (BoNT-A) treatment; conflict of interest, role of industry, and the role of history; and optimal treatment modalities and dose-response relationships. We conclude that several factors hindered acceptance of the Space Bop results. Awareness of these factors is important when performing rehabilitation research and disseminating and implementing research findings.

is consensus that BoNT-A is not a stand-alone intervention (1, 2). The rationale is that leg-muscle injected BoNT-A decreases spasticity by temporarily blocking neuromuscular transmission, which will provide better conditions for adjunctive rehabilitation interventions, and these are then assumed to be more effective. BoNT-A treatment is costly and places a burden (i.e. a toxin is injected, often preceded by some form of anaesthesia) on both the child and his or her caregivers.

Much research has been performed into the effects of BoNT-A treatment for the heterogeneous group of children with spastic CP. The overall conclusion is that BoNT-A has positive effects, e.g. on passive range of motion and on gait (1–7). However, in almost all studies one or more adjunctive interventions, besides BoNT-A, were part of the treatment, i.e. the adjunctive interventions were not explicitly withheld. The overall effects of treatment therefore result from the combination of BoNT-A *plus* adjunctive interventions. Until recently, the specific added value of the BoNT-A component as part of the treatment paradigm had not been investigated, although the need for studying the relative contribution of BoNT-A and other treatment components has been reported several times (5, 8–13).

This was the reason to initiate the SPAstic cerebral palsy: (Cost)Effectiveness of BOtulinum toxin and Physiotherapy (Space Bop) trial. Space Bop was designed

<sup>\*</sup>This article has been handled and decided upon by Chief-Editor Kristian Borg. Henk Stam has not been involved in the decision process.

as a comparative multi-centre cost-effectiveness trial from a societal perspective, in which "comprehensive rehabilitation only" was compared with "similar comprehensive rehabilitation, but preceded by BoNT-A". This design allowed conclusions to be drawn about the added value of the BoNT-A component of combined treatment for ambulatory children with lower limb CP. The study included a broad set of outcome measures across different domains of functioning (14), with the primary outcomes being gross motor function, everyday physical activity levels and health-related quality of life.

The Space Bop study was completed, and the comparative effectiveness results on the value of the BoNT-A injections preceding comprehensive rehabilitation have been published in this journal (15, 16). At first glance the results appeared to conflict with current evidence, but this conflict fades when the findings are placed in a broader context.

The current paper shares and discusses some noteworthy facts and experiences that we encountered while performing the Space Bop study, and while presenting, discussing and publishing the results. The paper describes and discusses some crucial aspects of the Space Bop study, related to the design and results, and their dissemination and implementation.

## SPACE BOP STUDY: DESIGN, RESULTS AND DISSEMINATION

## Design

Space Bop was funded by ZonMW (the Netherlands Organisation for Health Research and Development, ZonMW grant number 170995003, HTA funding call 2009), and was originally designed as a multicentre randomized clinical trial (n=60), in which treatment with comprehensive rehabilitation (only-CR) was compared with comprehensive rehabilitation preceded by BoNT-A injections (BoNT-A+CR). Primary outcomes from the perspective of rehabilitation medicine were gross motor function, everyday physical activity levels, and healthrelated quality of life, while the secondary outcomes included pain, motor impairments and gait kinematics. In addition, goal attainment scaling (GAS) was used, with goals defined in the activity domain. Data were analysed at primary end-point (3 months) and 6 months follow-up (another 3 months later) using linear mixed models.

During the study, full randomization appeared not to be feasible because of unanticipated, strong preferences of parents/caregivers for one of the 2 treatments (see Schasfoort et al. (15) for more extensive information). Therefore, approval from ZonMW and from the ethics committee of Erasmus MC (NL26738.078.09), the trial was registered in the Dutch Trial Register (Trial NL1576)., the study was altered to a pragmatically designed trial (eventual total 65 participants), in which 40% of the children with lower limb CP were randomized and 60% were assigned to the treatment arm their parents/caregivers preferred. This resulted in 41 children in the BoNT-A+CR group, and 24 in the only-CR group.

## RESULTS

The findings from Space Bop were clear and straightforward and have been described in 2 comparative effectiveness papers (15, 16): no differences between interventions were found at 3 and 6 months, i.e. with comprehensive rehabilitation there was no evidence for the added value of BoNT-A on any of the primary and secondary outcomes, nor on GAS. The total cost per child for treatment up to primary end-point at three months was significantly higher for BoNT-A+CR than for only-CR (8,963 vs 6,182 euro, p=0.001). Because of the pragmatic design, there were imbalances in size and some baseline patient characteristics between the (larger) group that received BoNT-A+CR and the (smaller) group that received only-CR. Because this could potentially affect the validity of findings, baseline comparability was described in detail (15, 16), and several additional analyses were performed that focused on the effects of potential sources of bias. These analyses did not indicate any confounding effect.

## Dissemination

After data analyses, the results were presented at several national and international conferences, and the 2 comparative effectiveness papers were submitted (15, 16). There was a lengthy (re)submission phase, during which many comments and, mainly design-related, immediate rejections were addressed. The fact that the study was not fully randomized was the main issue, but other methodological challenges known to occur in the evaluation of rehabilitation interventions, such as the effect of adjunctive interventions and expectation bias, were also topics of discussion (17).

In this period, both invitational and open-access project meetings were organized to inform children, parents, informal caregivers, medical specialists, therapists and other interested stakeholders. Especially during the meetings with clinicians, who were mainly rehabilitation specialists, 3 distinct groups were recognized: (*i*) believers: "I'm not surprised by the results, and I will change my treatment"; (*ii*) non-believers: "the results are not correct, especially given the methodological shortcomings (not being fully randomized)"; and (*iii*) the intermediate group: "the results might be correct, but there still might be reasons to apply BoNT-

Journal of Rehabilitation Medicine

A (e.g. for other subgroups with CP, other treatment goals), the indication for BoNT-A treatment could probably be narrowed".

## DISCUSSION

In short, the results of the Space Bop study gave rise to much discussion and, perhaps also, some confusion. The results were not automatically and immediately accepted and implemented. It is known that a general tendency exists to reject new knowledge, especially when it contradicts existing beliefs (known as the Semmelweis reflex), as described by Mortell and colleagues (18). It is also known that the implementation of scientific evidence and guidelines in clinical practice is a complex process, and that the determinants of a successful strategy are still not fully understood (19). Therefore, it might be expected that the acceptance of our counterintuitive results and discontinuation of the wide use of BoNT-A treatment would be a difficult and time-consuming process. What surely contributed to the scepticism we encountered was the fact that Space Bop was the first study to explicitly focus on the added value of BoNT-A as part of a multimodal intervention in the context of a single complete cycle of care, and the above-described methodological issues that we never dismissed. Nevertheless, we feel that other factors have played, and still play, a dominant role in the phase of acceptance and dissemination, as discussed below.

## Interpretation and presentation of previous research

Space Bop was the first study that specifically focused on the added value of BoNT-A preceding multimodal comprehensive rehabilitation in comparison with an equally intensive multimodal comprehensive rehabilitation intervention without BoNT-A, but it was not the first study on the effects of BoNT-A combined with comprehensive rehabilitation. As already elaborated on in the Introduction, the overall effects of treatment include the combination of BoNT-A plus adjunctive interventions (Fig. 1a). Strikingly, in most of the previous studies only the BoNT-A part was put forward as the effective component, thus disregarding the (potential) contribution of the modalities from the comprehensive rehabilitation part, such as physiotherapy, casting and orthosis (e.g. (5, 6, 20-24) (Fig. 1b). For instance, in the latest Cochrane review, all the comparisons made are presented as "BoNT-A" vs "other treatments" which does no justice to the actual comparison that has been made (4). As stated, Space Bop was the first study that aimed at this comparison (Fig. 1c).

When the literature is critically appraised, there are a number of indications that the relative contribution of factors other than BoNT-A to overall effectiveness of combined treatment packages may have been underestimated (2, 13, 22, 25–27). In short, a stronger relative contribution of factors other than BoNT-A may equally well explain improvements, especially in view of the longer-term effectiveness (16). It should be noted that the quality of the evidence for BoNT-A treatment of lower limb spasticity was low to very low in the most recent Cochrane review (4). In addition, from Norwegian registry data it has been reported that intensive training was the only intervention factor associated with enhanced gross motor progress (28, 29).



**Fig. 1.** (a) General paradigm: botulinum toxin combined with other treatment components results in beneficial effects. (b) General interpretation: beneficial effects are mainly resulting from botulinum toxin, thereby neglecting or diminishing effects of other components. (c) Space Bop study: *blue arrows* indicate the multimodal intervention including botulinum toxin; *red arrows* indicate the comparative intervention: all treatment components, but without botulinum toxin.

#### p. 4 of 6 J. B. J. Bussmann et al.

JRM

The main issue here is how results, almost without exception, have been presented in especially titles and abstracts of publications: namely as stand-alone BoNT-A effects. Such presentation is probably based on an implicit assumption that BoNT-A is the most active component in combined treatment packages (15), but does not match the comparison that has actually been made and is also not in line with the consensus that BoNT-A should be combined with other interventions in which a number of different clinicians is involved in a single treatment cycle. It has been reported that, in many cases, the interventions under comparison largely remained undefined black boxes (10, 15, 16, 30). This makes it unclear whether the firm conclusions regarding effectiveness of BoNT-A treatment have been drawn from a comparison of "apples and oranges" rather than from a fair comparison. For a proper interpretation of findings it is important to take this into account: presenting (differences in) effects (between groups) as stand-alone BoNT-A effects (especially when it is unclear whether children had specifically refrained from other interventions) may be quite misleading and may place readers of those papers, and possibly even reviewers or editors, on the wrong track.

# *The role of one's own clinical experience and interpretation*

In the Netherlands, rehabilitation specialists and neurologists lead the treatment of children with spastic cerebral palsy. After 1-day outpatient BoNT-A treatment in the hospital, the children visit a physiotherapist for a period of 3 months, and when indicated, they also receive casting and/or orthoses. Besides the phenomenon described above, it seems to be possible, or even logical, that medical specialists consider their role and contribution as the most important part of successful treatment. In the social psychology this is known as "self-serving bias" (31). Thus, when they notice or are informed about the positive effects of treatment, they might tend to experience their contribution as most important, especially in case they are not directly involved in the other (in this case mainly the physical therapy component) part of treatment. In contrast, when a negative effect is noted, it is tempting to attribute failure to others involved in the full cycle of care, or to refractory clinical practice in general. This mis-estimation or over-estimation of one's own role might have contributed to interpreting the Space Bop results as "counterintuitive".

## Shift in aims of (BoNT-A) treatment

During the course of the Space Bop study we observed interesting shifts in the discussion about the aims of treatment cycles. The a priori defined primary outcomes from the broad set of outcome measures were analysed first: gross motor function, everyday physical activity levels and health-related quality of life. When those results were negative regarding the added value of BoNT-A, "clinicians" (we are generalizing here) responded that those outcomes were not most important, that some of the secondary outcomes and GAS were more relevant. and that effects on these outcomes would be more logical given the BoNT-A working mechanism. When these latter outcomes (i.e. functional leg muscle, strength, passive range of motion, angle of catch, CP-related pain, walking speed, kinematic gait parameters, goal attainment, and proxy-reported general functioning) also showed negative results, it was reported that BoNT-A would certainly have other, long-term effects, such as the prevention of contractures, possibly only within a subgroup of ambulatory children with CP. We will not discuss the rationality of these assumptions, and BoNT-A might have relevant long-term effects, but we only want to emphasize the need of being precise on the initially defined research questions and outcomes, and not change the reasoning during the project. Being clear about the aims of treatment is also of major importance in clinical practice: children and their parents may expect to be precisely informed about the goals of treatment and the effects that can be expected.

## Conflicts of interest, role of industry, and history

Space Bop was funded by ZonMw, a Dutch governmental organization for health research and development. Space Bop therefore was not industry-initiated, and was completely independent of any commercial influence. There are indications that industry-initiated trials generally result in more positive (read: biased) results and conclusions (32). Even non-industry-initiated trials might be affected by dependencies between industry and medical specialists. In addition, industry-initiated trials generally have different goals; they focus on effect size (=large contrast between interventions), registration of a drug, and safety/efficacy. Therefore, as we described earlier, disentangling the relative contribution of the different modalities of treatment cycles is not of primary concern (16).

For a long time now, undoubtedly with positive intentions, the application of BoNT-A has been promoted, and, since the turn of this century, BoNT-A treatment is reimbursed by health insurance companies in the Netherlands, although paediatric use of BoNT-A other than for the gastrocnemii muscle complex is officially off-label (33). BoNT-A treatment is provided by trained and specialized rehabilitation specialists and/or neurologists, and it is imaginable that those medical experts are reluctant to abandon this intervention and experience the risk of having to abdicate their "professional child", especially since many feel it benefits the children.

## Optimal treatment and dose response

One of the critical comments on our study was related to the applied doses of BoNT-A, which were sometimes considered too low. However, the treatment modalities in the Space Bop study were carefully defined, and in agreement with the consensus of that moment (1, 2, ..., 2)34, 35), and, if needed, adapted by the latest scientific evidence. However, in the field of BoNT-A the number of dose-response studies is relatively limited and the available studies are difficult to interpret (7). Because of the challenges that are related to clinical trials in chronic (paediatric) rehabilitation populations, such as CP(13, 17, 36), for the future it is probably advisable to not only focus on randomized controlled trials, but also on studies that provide better insight into working mechanisms and dose-response relationships, which also account for the adjunctive (rehabilitation) interventions that are part of the full treatment cycle (4). This latter topic (i.e. the moderating effect of intensity and content of the different modalities applied, including the effect of guideline adherence for physical therapy periods) is the focus of a paper in preparation by our group.

Another discussion was about the mode of application of BoNT-A (using ultrasound, electro, manual as guidance); as long as the interplay of complex combined interventions is not taken into account in effectiveness research it is difficult to make a statement about this. Also, regarding recent BoNT-A safety discussions, it is important to always balance potential benefits of BoNT-A against possible negative long-term effects, such as muscle atrophy and fibrosis, and their subsequent implications for future functioning of children with CP (37).

### Conclusion

The conclusions of the Space Bop study, although we recognize the suboptimal methodology regarding randomization, were clear: in ambulatory children with CP there was no evidence for an added effect of leg-muscle injected BoNT-A prior to comprehensive rehabilitation on a large set of outcomes measures. However, acceptance of these findings appeared to be difficult. We do not question the working mechanisms of BoNT-A, nor do we want to overestimate the clinical implications of our work. Nevertheless, we think that the results should be recognized and should be considered carefully, with the topics of the current report in mind.

Implementing the results of scientific research is a societal responsibility of researchers, but this also appears to be complex and needs specific expertise. Barriers to this process can have a rational background, but it is important to be aware that irrational and undesirable barriers might be even more prominent. With the description and discussion of our Space Bop study as an example, we hope to contribute to elucidating the role and effect of these barriers.

## REFERENCES

- Heinen F, Desloovere K, Schroeder AS, Berweck S, Borggraefe I, van Campenhout A, et al. The updated European Consensus 2009 on the use of Botulinum toxin for children with cerebral palsy. Eur J Paediatr Neurol 2010; 14: 45–66.
- Love SC, Novak I, Kentish M, Desloovere K, Heinen F, Molenaers G, et al. Botulinum toxin assessment, intervention and after-care for lower limb spasticity in children with cerebral palsy: international consensus statement. Eur J Neurol 2010; 17 Suppl 2: 9–37.
- Ade-Hall RA, Moore AP. Botulinum toxin type A in the treatment of lower limb spasticity in cerebral palsy. Cochrane Database Syst Rev 2000(2): CD001408.
- Blumetti FC, Belloti JC, Tamaoki MJ, Pinto JA. Botulinum toxin type A in the treatment of lower limb spasticity in children with cerebral palsy. Cochrane Database Syst Rev 2019; 10: CD001408.
- Ryll U, Bastiaenen C, De Bie R, Staal B. Effects of leg muscle botulinum toxin A injections on walking in children with spasticity-related cerebral palsy: a systematic review. Dev Med Child Neurol 2011; 53: 210–216.
- Novak I, McIntyre S, Morgan C, Campbell L, Dark L, Morton N, et al. A systematic review of interventions for children with cerebral palsy: state of the evidence. Dev Med Child Neurol 2013; 55: 885–910.
- Guyot P, Kalyvas C, Mamane C, Danchenko N. Botulinum toxins type A (Bont-A) in the management of lower limb spasticity in children: a systematic literature review and Bayesian network meta-analysis. J Child Neurol 2019; 34: 371–381.
- Scholtes VA, Dallmeijer AJ, Knol DL, Speth LA, Maathuis CG, Jongerius PH, et al. The combined effect of lowerlimb multilevel botulinum toxin type a and comprehensive rehabilitation on mobility in children with cerebral palsy: a randomized clinical trial. Arch Phys Med Rehabil 2006; 87: 1551–1558.
- Koog YH, Min BI. Effects of botulinum toxin A on calf muscles in children with cerebral palsy: a systematic review. Clin Rehabil 2010; 24: 685–700.
- Sakzewski L, Reedman S, Hoffmann T. Do we really know what they were testing? Incomplete reporting of interventions in randomised trials of upper limb therapies in unilateral cerebral palsy. Res Dev Disabil 2016; 59: 417–427.
- Reedman S, Boyd RN, Sakzewski L. The efficacy of interventions to increase physical activity participation of children with cerebral palsy: a systematic review and meta-analysis. Dev Med Child Neurol 2017; 59: 1011–1018.
- Bjornson K, Hays R, Graubert C, Price R, Won F, McLaughlin JF, et al. Botulinum toxin for spasticity in children with cerebral palsy: a comprehensive evaluation. Pediatrics 2007; 120: 49–58.
- Multani I, Manji J, Hastings-Ison T, Khot A, Graham K. Botulinum toxin in the management of children with cerebral palsy. Paediatr Drugs 2019; 21: 261–281.
- World Health Organization (WHO). International Classification of Functioning. Geneva: WHO; 2001.
- Schasfoort F, Dallmeijer A, Pangalila R, Catsman C, Stam H, Becher J, et al. Value of botulinum toxin injections preceding a comprehensive rehabilitation period for children with spastic cerebral palsy: a cost-effectiveness study. J Rehabil Med 2018; 50: 22–29.
- Schasfoort F, Pangalila R, Sneekes EM, Catsman C, Becher J, Horemans H, et al. Intramuscular botulinum toxin prior to comprehensive rehabilitation has no added value for

#### p. 6 of 6 J. B. J. Bussmann et al.

improving motor impairments, gait kinematics and goal attainment in walking children with spastic cerebral palsy. J Rehabil Med 2018; 50: 732–742.

- 17. Wade DT, Smeets RJ, Verbunt JA. Research in rehabilitation medicine: methodological challenges. J Clin Epidemiol 2010; 63: 699–704.
- Mortell M, Balkhy HH, Tannous EB, Jong MT. Physician 'defiance' towards hand hygiene compliance: Is there a theory-practice-ethics gap? J Saudi Heart Assoc 2013; 25: 203–208.
- Mostofian F, Ruban C, Simunovic N, Bhandari M. Changing physician behavior: what works? Am J Manag Care 2015; 21: 75–84.
- Delgado MR, Tilton A, Russman B, Benavides O, Bonikowski M, Carranza J, et al. AbobotulinumtoxinA for equinus foot deformity in cerebral palsy: a randomized controlled trial. Pediatrics 2016; 137: e20152830.
- Dursun N, Bonikowski M, Dabrowski E, Matthews D, Gormley M, Tilton A, et al. Efficacy of repeat AbobotulinumtoxinA (Dysport(R)) injections in improving gait in children with spastic cerebral palsy. Dev Neurorehabil 2019 Nov 6 [Epub ahead of print].
- Molenaers G, Fagard K, Van Campenhout A, Desloovere K. Botulinum toxin A treatment of the lower extremities in children with cerebral palsy. J Child Orthop 2013; 7: 383–387.
- Lukban MB, Rosales RL, Dressler D. Effectiveness of botulinum toxin A for upper and lower limb spasticity in children with cerebral palsy: a summary of evidence. J Neural Transm (Vienna) 2009; 116: 319–331.
- 24. Garcia Salazar LF, dos Santos GL, Pavao SL, Rocha NA, de Russo TL. Intrinsic properties and functional changes in spastic muscle after application of BTX-A in children with cerebral palsy: systematic review. Dev Neurorehabil 2015; 18: 1–14.
- Reddihough DS, King JA, Coleman GJ, Fosang A, McCoy AT, Thomason P, et al. Functional outcome of botulinum toxin A injections to the lower limbs in cerebral palsy. Dev Med Child Neurol 2002; 44: 820–827.
- Love SC, Valentine JP, Blair EM, Price CJ, Cole JH, Chauvel PJ. The effect of botulinum toxin type A on the functional ability of the child with spastic hemiplegia a randomized controlled trial. Eur J Neurol 2001; 8 Suppl 5: 50–58.

- 27. Yap R, Majnemer A, Benaroch T, Cantin MA. Determinants of responsiveness to botulinum toxin, casting, and bracing in the treatment of spastic equinus in children with cerebral palsy. Dev Med Child Neurol 2010; 52: 186–193.
- Storvold GV, Jahnsen RB, Evensen KAI, Bratberg GH. Is more frequent physical therapy associated with increased gross motor improvement in children with cerebral palsy? A national prospective cohort study. Disabil Rehabil 2018 Nov 16 [Epub ahead of print].
- 29. Storvold GV, Jahnsen RB, Evensen KAI, Romild UK, Bratberg GH. Factors associated with enhanced gross motor progress in children with cerebral palsy: a register-based study. Phys Occup Ther Pediatr 2018; 38: 548–561.
- Ryan JM, Cassidy EE, Noorduyn SG, O'Connell NE. Exercise interventions for cerebral palsy. Cochrane Database Syst Rev 2017; 6: CD011660.
- Mezulis AH, Abramson LY, Hyde JS, Hankin BL. Is there a universal positivity bias in attributions? A meta-analytic review of individual, developmental, and cultural differences in the self-serving attributional bias. Psychol Bull 2004; 130: 711–747.
- 32. Sung KH, Chung CY, Lee KM, Lee YK, Lee SY, Lee J, et al. Conflict of interest in the assessment of botulinum toxin A injections in patients with cerebral palsy: a systematic review. J Pediatr Orthop 2013; 33: 494–500.
- Nahm NJ, Graham HK, Gormley ME, Jr., Georgiadis AG. Management of hypertonia in cerebral palsy. Curr Opin Pediatr 2018; 30: 57–64.
- Verschuren O, Balemans AC. Update of the core set of exercise tests for children and adolescents with cerebral palsy. Pediatr Phys Ther 2015; 27: 187–189.
- Verschuren O, Peterson MD, Balemans AC, Hurvitz EA. Exercise and physical activity recommendations for people with cerebral palsy. Dev Med Child Neurol 2016; 58: 798–808.
- Kersten P, Ellis-Hill C, McPherson KM, Harrington R. Beyond the RCT – understanding the relationship between interventions, individuals and outcome – the example of neurological rehabilitation. Disabil Rehabil 2010; 32: 1028–1034.
- Multani I, Manji J, Tang MJ, Herzog W, Howard JJ, Graham HK. Sarcopenia, cerebral palsy, and botulinum toxin type A. JBJS Rev 2019; 7: e4.